Author's response to reviews

Title: Difference of the associations between self-rated health and demographic characteristics, lifestyle, and psychosocial work environment between two representative types of Chinese worksite

Authors:

Yingnan Jia (12111020010@fudan.edu.cn)
Junling Gao (jlgao@fudan.edu.cn)
Junming Dai (jmdai@shmu.edu.cn)
Pinpin Zheng (zpinpin@shmu.edu.cn)
Xiaoyu Wu (wuxiaoyu@shawh.gov.cn)
Guangyao Li (liguangyao@shawh.gov.cn)
Hua Fu (hfu@fudan.edu.cn)

Version: 5

Date: 6 August 2014

Author's response to reviews:

Response to Reviewer

Title: Difference of the associations between self-rated health and demographic characteristics, lifestyle, and psychosocial work environment between two representative types of Chinese worksite

Reviewer's report:

1. The authors must be consistent in using the term "type of work" and "types of work."

Response to Reviewer's comments:

The instances of type vs. types of work have been revised.

2. This sentence "Self-rated health (SRH) is generally assessed using a single survey..." sounds confusing..."

Response to Reviewer's comments:

It would be better to say: SRH is generally assessed by a single item index or a single survey question.

3. More editorial assistance is needed...

Response to Reviewer's comments:

The manuscript has received a professional editing service. We sincerely hope this manuscript will be finally acceptable to be published on BMC Public Health.